Author's response to reviews

Title: Water intoxication presenting as maternal and neonatal seizures: a case report

Authors:

Timothy H Chapman (timothy.chapman@stgeorges.nhs.uk)
Mark Hamilton (markhamilton@nhs.net)

Version: 3 Date: 3 August 2008

Author's response to reviews:

Subsequent to the two reviewer's reports, I have edited the manuscript as follows;

The patient's age in the abstract has been changed to 37 years as this was her presenting age.

The statement regarding cause of osteoporosis has been altered to - 'The osteoporosis could be associated with the pregnancy, as no other cause was identified in this young primigravid woman.' Thus suggesting the possibility of pregnancy associated osteoporosis rather than attributing this to being the cause.

The reason for intubation has also been clarified as being due to decreased consciousness due to intravenous anti-convulsant therapy.

The paragraph 2, page 6 has also been re-phrased to prevent confusion - 'This rapid correction was done due to her ongoing seizure risk and stopped when the serum sodium reached 125-130mmol/l (Table 1) or the patient deemed to no longer be at risk of life-threatening manifestations of severe hyponatremia with cessation of seizure activity.'

Finally the time period has been corrected in page 6 paragraph 2 to 12mmol/l per day.